Author's response to reviews

Title: A large community outbreak of waterborne giardiasis- delayed detection in a non-endemic urban area

Authors:

Karin Nygard (kany@fhi.no)
Barbara Schimmer (basc@fhi.no)
Oystein Sobstad (oysteinsobstad@gmail.com)
Anna Karen Walde (Anna.Karen.Walde@mattilsynet.no)
Ingvar Tveit (Ingvar.Tveit@bergen.kommune.no)
Nina Langeland (nina.langeland@helse-bergen.no)
Trygve Hausken (trygve.hausken@helse-bergen.no)
Preben Aavitsland (praap@fhi.no)

Version: 2 Date: 22 February 2006

Author's response to reviews:

We were asked to add a sentence about ethical approval. We added an explanation in the methods section.